
    PS nTOF 

Facility Coordinator last week Michael Bacak 

Facility Coordinator this week  

Beam Requested 

EAR 1 No EAR 2 No 

Facility Status 

Summary 
• DSO tests in n_TOF (including TT2A) are completed (23/3/2023).   

• Cable re-arrangement in EAR1 and EAR2 towards improved 
EM compatibility finalized. 

Issues - 

Plans 

• Ramp up of n_TOF target cooling/moderation circuits (STI) 

• EAR1: det./antennas - investigation of “ringing” problem (=5/2.5 MHz signal 
oscillation just after the gamma flash) 

• EAR1 neutron escape line: finalize modifications and vacuum tests 

• NEAR: setup of infrastructure for diamond detector flux measurement (xy-
table, additional shielding) 

• Finalize fine tuning of FEC + new DAQ for NEAR (EAR3) 

Foreseen Beam Stop 

Yes Duration - Date/Time - 

 


